Author's response to reviews

Title: A population-based survey on tanning bed use in Germany

Authors:

Franziska U Börner (f.boerner@fz-juelich.de)
Holger Schütz (h.schuetz@fz-juelich.de)
Peter Wiedemann (p.wiedemann@fz-juelich.de)

Version: 3 Date: 28 May 2009

Author's response to reviews: see over
Dear Mr. Kerr,

on behalf of my co-authors and myself, I inform you that we have revised the manuscript to address the comments from the reviewers detailed in your email from May 18, 2009.

To assist you in the final consideration of the manuscript, I have attached a file containing the “Responses to the Comments” that details the reviewers’ comments.

I thank you for your consideration to possibly publish our manuscript and hope that you agree that we have adequately addressed all comments. On behalf of the authors, I like to thank you and the reviewers for all your much appreciated efforts with our manuscript.

Sincerely,

Franziska Börner

Institute of Neurosciences and Medicine
INM-8: Ethics in the Neurosciences
Forschungszentrum Jülich
52425 Jülich
Germany
Response to Editor & Reviewer Comments for Manuscript ID MS: 1528865924237469 entitled “A population-based survey on tanning bed use in Germany.”

<table>
<thead>
<tr>
<th>Nr.</th>
<th>Comment</th>
<th>Response</th>
<th>Authors’ Decision</th>
</tr>
</thead>
<tbody>
<tr>
<td>1</td>
<td>Go through the manuscript formatting checklist one more time and ensure that your revised manuscript conforms to all of the points.</td>
<td>Agree. We have revisited the journal’s Instructions to Authors in regards to the journal style and made appropriate changes.</td>
<td>We have deleted “et al.” in one reference and added all additional authors, see references.</td>
</tr>
</tbody>
</table>

**Reviewer # 1**

1. No comments

**Reviewer # 2**

1. The article can be accepted if the table 4 and Figure 1 are quoted IN the text or if they are dropped out of the article.

Table 4 and Figure 1 are quoted appropriately in the text (see Table 4 page 9, line 14; Figure 1 page 8, line 20). We do, however, acknowledge that due to the text length more mentioning of Table 4 and Figure 1 would be appropriate to accommodate the general understanding.

We have added the phrases “Figure 1”, see page 9, line 12 “Table 4”, see page 9, line 17